

Intrapreneurship activity and access to finance in natural science: evidence from the UK academic spinoffs

Article

Published Version

Creative Commons: Attribution 4.0 (CC-BY)

Open Access

Audretsch, D. B., Belitski, M. ORCID: https://orcid.org/0000-0002-9895-0105 and Scarra, D. (2024) Intrapreneurship activity and access to finance in natural science: evidence from the UK academic spinoffs. Technovation, 129. 102888. ISSN 1879-2383 doi:

https://doi.org/10.1016/j.technovation.2023.102888 Available at https://centaur.reading.ac.uk/119242/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1016/j.technovation.2023.102888

Publisher: Elsevier

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other copyright holders. Terms and conditions for use of this material are defined in the <u>End User Agreement</u>.

www.reading.ac.uk/centaur



CentAUR

Central Archive at the University of Reading

Reading's research outputs online

Contents lists available at ScienceDirect

Technovation

journal homepage: www.elsevier.com/locate/technovation

Intrapreneurship activity and access to finance in natural science: Evidence from the UK academic spinoffs

David B. Audretsch^a, Maksim Belitski^{b,c,d,*}, Deepa Scarra^e

^a School of Public and Environmental Affairs, Indiana University Bloomington, 1315 E. 10th Avenue SPEA, Bloomington, IN, 47405, USA

^b Loyola University New Orleans, College of Business, USA

^c ICD Business School, IGS-Groupe, France

^d Henley Business School, University of Reading, United Kingdom

^e Sant'Anna School of Advanced Studies, Piazza Martiri della Libertà, 33, 56127, Pisa, Italy

ARTICLE INFO

Keywords: Natural sciences Intrapreneurship University Spinoffs Digitally driven Knowledge transfer

ABSTRACT

Recent research on the intrapreneurial activity at universities has demonstrated that access to finance and firm profits highly depend on the adoption of digital technologies and the stage of growth. This paper aims to contribute to the ongoing debate on the factors that affect access to finance and profitability of academic spinoffs across different stages of spinoff growth. We distinguish and define an intrapreneurial academic spinoff to argue that the intrapreneur's own development of digital technology, stage of growth, and the field of science represent three boundary conditions to access external finance and gain profitability. Using the data from 89 intrapreneurial academic spinoffs in natural sciences, such as physical sciences and astrophysics, mathematics, chemical sciences, earth and related environmental sciences as well as 660 spinoffs from other fields of science in the United Kingdom over 2015–2019, this paper examined the role of spinoff growth stage and development of digital technologies in access to finance and profitability of this type of spinoffs. The paper offers implications for policymakers, intrapreneurial spinoffs, and university managers.

1. Introduction

The landscape of universities has recently undergone a notable shift towards intrapreneurial activities, fuelled by the commercialization of latent academic knowledge research. This trend has gained significant attention in the realms of technology entrepreneurship and ecosystems research (Bienkowska and Klofsten, 2012; Bienkowska et al., 2016; Brem and Radziwon, 2017; Klofsten et al., 2021). At the core of this shift is the academic scientist, whether they play the role of a researcher or teacher, driven by the goal to act entrepreneurially (Vohora et al., 2004; Abreu and Grinevich, 2013; Guerrero and Urbano, 2012). Their objective is clear: to harness and bring to market the new knowledge they generate, which, if uncommercilized, remains unexploited (Van Burg et al., 2008).

However, the landscape of entrepreneurship within universities is not homogeneous. While some researchers generate basic knowledge (Audretsch, 2014) and stay within the realm of university, others focus on applied knowledge and combining their research roles with

entrepreneurial activity giving birth to academic spinoffs, thus embodying the role of 'intrapreneurs' (Klofsten and Jones-Evans, 2000; Klofsten, 2008; Vanhaverbeke and Peeters, 2005). As defined by Baruah and Ward (2015: 811-812), intrapreneurship activity is "the innovation practice within an organization through which employees undertake new business activities and pursue different opportunities". A similar understanding of intrapreneurship is echoed by Burkholder and Hulsink (2022: 1), stating that "academic intrapreneurship refers to the individual behaviors of scientists who depart from their customary research and education initiatives and become involved in knowledge commercialization without leaving academia". These individuals who leave their professional role at the university to pursue entrepreneurial opportunities are entrepreneurs (Parker, 2011). Their ventures are often rooted in the knowledge spillover theory of entrepreneurship, which posits that uncommercialized knowledge, typically in a foirm of basic knowledge created within a university, can be commercialized by intrapreneurs (Guerrero and Urbano, 2014). These academic entrepreneurs breathe life into basic knowledge, either by initiating new firms or by creating academic spinoffs (Stuart and Ding,

https://doi.org/10.1016/j.technovation.2023.102888

Received 6 October 2022; Received in revised form 29 August 2023; Accepted 1 October 2023 Available online 5 October 2023





 $^{^{\}ast}$ Corresponding author. Loyola University New Orleans, College of Business, USA.

E-mail addresses: daudrets@indiana.edu (D.B. Audretsch), mbelitsk@loyno.edu, mbelitski@groupe-igs.fr, m.belitski@reading.ac.uk (M. Belitski), deepa.scarra@ santannapisa.it (D. Scarra).

^{0166-4972/© 2023} The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

2006; Visintin and Pittino, 2014; Fukugawa, 2022; Burkholder and Hulsink, 2022). University often remains a key stakeholder in academic spinoffs and retains spin-off shares.

Despite considerable advancements in research on intrapreneurial capabilities within companies and universities (Klofsten et al., 2021; Guerrero et al., 2016, 2021) and the antecedents of university intrapreneurship (Valka et al., 2020), gaps remain in understanding the key mechanisms and boundary conditions that allow academic spinoffs to commercialize their knowledge and achieve high performance, their growth trajectory, and the timeline for fundraising and growth. Existing research has also not adequately explored the factors and conditions that might either bolster or hinder the visibility and hence the ability of spinoffs to use different sources of funding for growth (Åstebro et al., 2013, 2019; Civera et al., 2020; Di Gregorio and Shane, 2003).

Access to finance and profitability are two interrelated challenges that are especially pronounced in the early stages of spinoff growth e.g. pre-trading and early market entry (Clarysse et al., 2007). At these phases, investors typically seek tangible evidence of potential success, like a minimum viable product or a proof of concept, before committing funds to either basic or applied research (Shane and Stuart, 2002; Conti et al., 2013; Bradley et al., 2013; Abreu and Grinevich, 2013). Thus, academic spinoffs often grapple with substantial obstacles in fundraising and profit generation if unable to expedite experimentation and prove their concept at early stages. While academic spinoffs that fail to showcase market-ready products and investment in digital technologies significantly hamper their signaling ability to investors, subsequently affecting their propensity to fundraise and profitability (Guerrero et al., 2015; Cunningham et al., 2020), the boundary conditions that shape propensity of fundraise and firm performance remain unknown. Field of science is an impactful moderator of this relationship.

In the natural science field, the journey from invention to commercialization is lengthy and is fraught with high market and product uncertainties, risks associated with experimentation, and challenges of adopting new and complex digital technologies. Spinoffs in natural sciences, as opposed to other fields of science, embody the fundamental disciplines and academic traditions of the Humboldt University model. Researchers in natural sciences prioritize "knowledge for its own sake," which Audretsch (2014: 317) recognizes as the gold standard of scholarly inquiry under the Humboldt University model, rather than seeking applied solutions to pressing societal problems and challenges. Consequently, the nature of knowledge produced in natural sciences differs substantially from the applied knowledge generated in fields such as biochemistry, informatics, and bioengineering. These fields are tasked with devising solutions and applications for significant societal challenges.

Given the unique characteristics of natural science spinoffs that hinge on basic knowledge, intrapreneurs often encounter immense challenges in procuring finance during the early stages and often invest in digital technologies to accelerate the journey of knowledge transfer. Without the commitment of industry and government to invest significant funds to transform a fundamental research idea into a market-ready product, the transition from basic knowledge creation to experimentation, fundraising, market commercialization, and initial profitability in the natural sciences is notably protracted compared to other scientific domains (Audretsch et al., 2023).

We aim to fill this existing gap in the literature by asking the following research question: "What is the role of boundary conditions such as the stage of growth, digitization, and the field of science in influencing access to external finance and profitability for intrapreneurial academic spinoffs?" To answer this question, we consider both external factors (e.g., field of science) and internal ones (e.g., stage of growth and investment in technology) that might either facilitate or hinder fundraising and profitability (Mustar et al., 2006; Visintin and Pittino, 2014; Li et al., 2016; Connected Commerce Council, 2020, 2021).

We contribute to the academic entrepreneurship literature by demonstrating the extent to which field of science, adoption of digitallyadvanced technologies and stage of growth as three boundary conditions can facilitate the propencity to fundraise from various sources and profitability in intrapreneurial academic spinoffs. We do so by employing the firm growth (Churchill and Lewis, 1983; Vohora et al., 2004) and digital capabilities literatures (Giones and Brem, 2017; Secundo et al., 2020; Jovanovic et al., 2021), and by distincting between digitally-advanced and digitally-uncertain intrapreneurial spinoffs (Connected Commerce Council, 2020, 2021).

We contribute to intrapreneurship literature calling for a deeper understanding of the differences between intrapreneurship and entrepreneurship within a university (Urbano et al., 2013; Meoli and Vismara, 2016; Klofsten et al., 2019), by examining how the development of digital technologies and capabilities as well as the timeline of growth changes the propensity to fundraise and grow profits between intrapreneurial spinoffs in natural and non-natural sciences field.

Our findings provide key insights to practitioners, policymakers and university managers on better support and manage natural science intrapreneurial spinoffs at universities. We appeal to policymakers to better support academic intrapreneurship in natural science and learn to recognize important signals from intrapreneurial spinoff in this field (Conti et al., 2013). Drawing on Klofsten et al. (2021) in particular, we argue that the development of digital technology in-house by an intrapreneurial spinoff in the natural sciences provides an additional positive signal to investors attracting more financial equity and debt capital and increasing financial performance (profits). While little attention has been paid to the role of digitalization in intrapreneurial spinoffs – spinoffs that develop and use new digital technology and find it essential in their business model - are able to access more financial resources and are able to do so earlier than their digitally-uncertain counterparts.

The remainder of the paper is organized as follows. Section 2 discusses the theoretical framework, and Section 3 develops the main hypotheses. Section 4 describes the data and the model specifications used in the empirical analysis, while Section 5 reports the empirical findings. Section 6 discusses the main findings and implications, while Section 7 concludes with theoretical development, limitations, and future research.

2. Theoretical framework

2.1. Intrapreneurship and entrepreneurship activity at university

In contrast to entrepreneurship activity at a university aiming to commercialize the outcomes of university research (Abreu and Grinevich, 2013; Audretsch, 2014), intrapreneurship activity takes place within a university and in the form of academic spinoffs (Jones-Evans and Klofsten, 1997; Klofsten et al., 2021). While some intrapreneurial spinoffs may be physically located outside of the university campus, they still belong to and are an integral part of the university if they are owned, co-owned, and managed by the university board and if they are related to university staff in their professional and/or research position. Intrapreneurial academic spinoffs are often small firms (Colombo et al., 2019) operating in high-tech and emerging industries (Bock et al., 2018; Visintin and Pittino, 2014) with high human capital (Unger et al., 2011), usually founded by small entrepreneurial teams (Aspelund et al., 2005). Drawing on prior research by Covin and Slevin (1991), recent research by Klofsten et al. (2021: 2) poses that "Intrapreneurship involves a company extending its competence and increasing its opportunities by creating new organizations, new products/services- or combining new resources". This is consistent with prior works of Antoncic and Hisrich (2003), Parker (2011), Guerrero et al. (2016) and Perlines et al. (2022), that entrepreneurship is known as the act of developing a new venture outside of an existing organization, while intrapreneurship activity is the process of developing a new venture within an existing organization.

That said, there is still no consensus about the dimensions of intrapreneurship or its definition, determinants and conditions (Farrukh et al., 2017; Neessen et al., 2019). The leading role of university employees at intrapreneurial spinout has been emphasized by Neessen et al. (2019: 545) who describe a "comprehensive model of intrapreneurship in which we integrate the new definition, dimensions, and determinants applicable to individual employees". Authors find that innovativeness, proactiveness, risk-taking, access to finance, opportunity recognition/exploitation and internal/external networking are important factors explaining the choice of intrapreneurial activity.

To promote both intrapreneurship and entrepreneurship in universities, management focuses on nurturing intrapreneurial abilities (Klofsten et al., 2021) to fortify such behavior amongst university researchers (Neessen et al., 2019; Cunningham et al., 2022). This commitment not only enhances the scientific quality but also elevates the likelihood of university spinoffs securing investments from venture capitalists, business angels, and other financial entities (Fukugawa, 2022). Scholars also define an intrapreneurial academic spinoff as "a process where by employees recognize and exploit opportunities by being innovative, proactive and by taking risks, in order for the organisation to create new products and services, initiative self-renewal or venture new businesses to enhance the competitiveness and performance of the organization" (Neessen et al., 2019: 551). Thus, intrapreneurial academic spinoff is a new venture established within a university with a founder (co-founder) or a CEO or both are employees at the university who apply their research in recognition of entrepreneurial opportunities, taking a pro-active innovative approach and taking calculated risks and for the university and a spinoff to create new products and services and commercialize them on the market.

The condition of university staff to be associated with the academic spinoff in a position of a (co)founder or a CEO, enables us to categorize a spinoffs as intrapreneurial academic spinout, rather than an entrepreneurial firm (Civera et al., 2020). Establishing an intrapreneurial spinoff triggers intrapreneurial behavior and activities among university employees, encouraging knowledge commercialization among university faculty and staff (Shane, 2004; Wright et al., 2006).

A predominant barrier for both intrapreneurship and entrepreneurship is resource availability (Klofsten et al., 2019), especially financial resources during the nascent stages of product development and market introduction. This makes intrapreneurial spinoffs rely substantially on university knowledge, networks, financial and infrastructure support (Soetanto and Van Geenhuizen, 2015), as well as support in accessing external finance such as grants, debt and equity funding (Clarysse et al., 2007; Munari and Toschi, 2011). Oftentimes scientists opt to retain their roles within the university, focusing on intrapreneurial activities instead of starting a new business. Supporting these initiatives can augment the university's value and facilitate access to external funding, catalyzing growth in academic spinoffs (Urbano et al., 2013). In pursuit of an intrapreneurial activity, spinoffs will seek three different sources of funding such as debt finance (e.g. short and long term bank loans) (Brown and Lee, 2019), equity finance (equity-based crowdfunding, venture capital and business angels) (Knockaert et al., 2010; Colombo et al., 2022) as well as innovation competitions and grant funding. Grant funding may be most preferred form of financing at the early stage of an intrapreneurial spinoff as this does not require dilution of equity or financial liabilities such s bank loans and paying interest rates (Gustafsson et al., 2020). The recent study of Alsos and Ljunggren (2017) using the signaling theory has demonstrated that positive signals to investors may include a firm having a prototype or already invested in R&D and digital capabilities could increase the likelihood of equity fundraising or receiving the bank loan, while academic spinoffs who received grants were also more likely to secure venture capital (Belz et al., 2021).

2.2. Access to finance and digitalization in intrapreneurial spinoffs

There is limited knowledge regarding the varied ways in which intrapreneurial academic spinoffs are established, and their strategies for securing financial resources and growth, especially across different stages of the growth continuum and various scientific fields. For instance, Bienkowska et al. (2016) showed that academics in the fields of natural science and technology often see the creation of a spinoff company as the primary outcome of entrepreneurial activity. In contrast, those in the arts and social sciences are more inclined towards the social dimensions of entrepreneurship. For them, organizing activities that address societal challenges is a more significant aspect of entrepreneurship (Bienkowska et al., 2016).

Despite the importance of natural science intrapreneurial spinoffs to universities, regions, and society at large, few studies have delved into explaining how these spinoffs secure funding and achieve greater profitability (e.g., Clarysse et al., 2011; Mathisen and Rasmussen, 2019). Compared to spinoffs in other scientific disciplines, intrapreneurial academic spinoffs in natural sciences may confront higher risks due to the intricate technologies involved and the experimental nature of their product innovations and commercialization processes (Clarysse et al., 2005; Moray and Clarysse, 2005). Given the long-term nature of research outcomes, the technological and market uncertainties inherent in natural science innovations, and especially in the early (pre-market) stages, these spinoffs might face challenges in fundraising and accessing bank capital (Soetanto and Van Geenhuizen, 2015). Often, these spinoffs introduce innovative products that need further validation and testing, pushing back their market entry timelines (Colombo et al., 2010; Sørheim et al., 2011). As a result, there's typically a substantial delay for natural science intrapreneurial spinoffs between academic discovery (Gruber et al., 2013), innovation commercialization, and the mass production phase (Wright et al., 2006; Clarysse et al., 2011; Knockaert et al., 2011). This elongated timeline hampers their ability to demonstrate profitability and secure financing to potential investors (Moray and Clarysse, 2005). Therefore, it is important to understand the factors and boundary conditions that may enhance or impede access to finance and profitability of intrapreneurial spinoffs (Vincett, 2010).

3. Hypothesis development

3.1. Intrapreneurial spin-offs and access to finance

Intrapreneurial activity within universities is diverse (Mustar et al., 2006). Heirman and Clarysse (2004) as well as Klofsten and Jones-Evans (2000) examined various academic spinoffs by considering the unique configurations of financial, technological, and human resources. Specifically, Heirman and Clarysse (2004) investigated how variations in a spinoff, combined with environmental factors like technological domain, organizational origin, and industry, can influence resource configuration and performance outcomes. In terms of technological resources, they took into account the innovativeness of the company's core technology, the development stage of the product/technology at the company's inception, and the scope of the product/technology—distinguishing between companies that were developing a singular product and those creating a platform for multiple products.

To manage the high uncertainty and risks associated with access to infrastructure, financial and human resources, the founder or CEO of a spinoff often maintains a university position. This provides a steady income, access to scientific labs and materials, doctoral students, an experimental base, and other university-wide support mechanisms such as technology transfer offices (TTO), university-industry partnerships, and science parks (Audretsch and Belitski, 2019). This foundational support promotes a higher tolerance for risk-taking in uncertain situations.

Although universities frequently back intrapreneurial spinoffs with resources, investors, lenders, and public resource providers might view these spinoffs as too uncertain or risky to finance (Druilhe and Garnsey, 2004). This perspective is especially prevalent for academic spinoffs in natural sciences, for several reasons.

Firstly, the founder or CEO is often only partially committed to the new venture, and the university maintains administrative oversight over the spinoff. Not all investors have experience collaborating with universities, and only a minority of them have worked with academic spinoffs in natural sciences. These spinoffs face an increased information asymmetries,

because it is often difficult to gauge technology and product maturity in the realm of natural sciences deadling wwith basic knowledge (Gompers, 1995). Secondly, innovations from intrapreneurial spinoffs in the natural sciences demand more time for market validation, knowledge testing, and the experimental phase. This often results in a prolonged commercialization timeline from invention to market, compared to spinoffs in other fields of science (Colombo et al., 2010; Sørheim et al., 2011). Thirdly, due to their technological complexity, academic spinoffs in natural sciences take longer to transition from the research environment to the marketplace (Heblich and Slavtchev, 2014). These entities require specialized equipment and materials for both experimentation and product development and fundamental research-this might include ordering unique equipment and testing materials such as chemical compounds, magnetic accelerators, microscopes, cyclotrons, telescopes and other. Finally, the university usually retains ownership of the intellectual property and has a say in decision-making processes (Sørheim et al., 2011; Munari and Toschi, 2011). To consider access to finance in the different stages of development of intrapreneurial spinoffs, we draw on the early stages of the firm growth model (Churchill and Lewis, 1983), distinguishing between the seed stage, venture/early growth stage, growth stage, established stage, exit stage, and dead stage. Based on the above argument, we hypothesize.

H1a. Intrapreneurial academic spinoffs in natural sciences fundraise equity and debt capital at later stages of firm growth compared to intrapreneurial academic spinoffs in other fields of science.

Most intrapreneurial spinoffs in natural science take a long period to create and transfer knowledge, as intrapreneurial capabilities take longer than other capabilities to develop, integrate, and coordinate, so investing in natural science start-ups creates uncertainty for investors and affects early sales and profits (Bradley et al., 2013).

Guerrero et al. (2020, 2021) explored the relationships between the ordinary capabilities needed to fulfill a university's core strategies (i.e., teaching quality, research quality, and administration) and intrapreneurial capabilities essential for executing a university's entrepreneurial strategy (like sensing opportunities and transforming routines to foster innovation). Their research also considered university outcomes. The speed of developing, integrating, and coordinating skills/assets/-knowledge transfer hinges on an intrapreneurial spinoff's aptitude for acquiring external knowledge or nurturing it internally, processes that are time-intensive. Intrapreneurial capabilities are essential for commercializing innovation, necessitating the genesis of novel technological or innovative ideas before a spinoff can seize economic value in the market – i.e., earn profits.

Bradley et al. (2013) posited that when technology is in its nascent stages, as is often the case with endeavors initiated by natural scientists, it appears too high-risk to entice investors or secure loans from banks. This perception persists unless the technology has been rigorously tested and validated (Colombo et al., 2010), processes that entail laboratory experiments, fundraising efforts, initial market sales, and the generation of preliminary profits. An intrapreneurial academic spinoff might be the sole avenue for developing and commercializing such technology. The formation of a spinoff can make the technology commercially viable (Shane, 2004). Rival researchers might attempt to contest these technologies both formally and informally, potentially influencing pre-market entry and future profit trajectories. Such challenges make investors and lenders particularly circumspect during the early stages of technology development.

Intrapreneurial academic spinoffs in natural science will need more time to claim intellectual property rights (Teixeira and Ferreira, 2019), collaborate on technology with potential customers, suppliers and competitors (Audretsch and Belitski, 2020, 2023), and facilitate transfer technology (Chiesa and Piccaluga, 2000) to secure its property rights, develop products and enter the market. Investors and banks would wish to observe product development and granting intellectual property rights and first profits before investing in the spinoff or lending financial resources. Investors and financial institutions typically prefer to monitor product evolution, the awarding of intellectual property rights, and the realization of initial profits before committing resources to the spinoff. This approach stands in contrast to intrapreneurial academic spinoffs in other scientific fields, where research outcomes might be more immediately discernible and market-ready (e.g., in domains like food science research, food manufacturing, construction, architecture, design, and agriculture). Natural science-focused intrapreneurial academic spinoffs are likely to generate profits only after market entry, and this profitability needs validation by financial stakeholders. Such dynamics slow the growth trajectory of intrapreneurial spinoffs, impacting their fiscal performance in the spinoff's nascent stages. As a consequence, profits are deferred from the venture and growth phases to the more mature stages of the spinoff's lifecycle. Based on this we hypothesize.

H1b. Intrapreneurial academic spinoffs in natural sciences increase their profits at later stages of firm growth compared to intrapreneurial academic spinoffs in other fields of science.

3.2. Intrapreneurial spin-offs and digitalization

Entrepreneurship and innovation scholars have recognized the role of digitalization as a means to develop intrapreneurial capabilities (Guerrero et al., 2021; Rippa and Secundo, 2019), with digital technologies becoming enablers of intrapreneurship at universities (Rabl et al., 2022). Digital capabilities are an integral part of intrapreneurial capabilities (Guerrero et al., 2021) and facilitate the development and adoption of digital technologies by intrapreneurial spinoffs. Spinoffs that develop digital technologies, invest in digital capabilities, and consider them as part of their business models are defined as digitally driven spinoffs (Connected Commerce Council, 2020). These digitally driven spinoffs are more resilient than digitally uncertain spinoffs. They create more jobs, attract more capital investment, and achieve higher profits. In contrast, intrapreneurial spinoffs that neither develop nor invest in digital technologies, and do not view digital capabilities as essential to their business model, are classified as digitally uncertain spinoffs. Such spinoffs are less resilient, more likely to experience slower growth rates, create fewer jobs, and access less capital (Connected Commerce Council, 2021). Within universities, intrapreneurial spinoffs leverage digital capabilities and develop digital technologies to explore and exploit market opportunities (Giones and Brem, 2017; Secundo et al., 2020). Digital tools are ubiquitous (Li et al., 2016) and can be employed by intrapreneurial academic spinoffs across different fields to enhance processes and operations and to innovate new products and services (Audretsch and Belitski, 2021a). Such technologies can diminish the costs associated with knowledge transfer and operations, signaling quality to stakeholders like investors, governments, and customers (Radko et al., 2022).

The reasons for the dependence of digitally-driven intrapreneurial spinoffs, particularly in natural sciences, on digital technologies in their business models to generate economic value and secure financing are manifold:

Firstly, Digital technologies enhance visibility on websites, enable targeted advertising for new products, and facilitate profiles on digital platforms such as crowdfunding and venture capital platforms. This increases access to equity and debt resources crucial for technology investment and growth (Bock et al., 2018; Knockaert et al., 2011). Spinoffs showcasing evidence of in-house digital technology development or digital tech incorporation that disrupts the market are more likely to secure external funds and boost sales (Belitski and Boreiko, 2022).

Secondly, incorporating digital tech results in better servitization, which is otherwise limited in natural science products. This transformation refines spinoff business models, aligning business with IT strategies and thus optimizing operations for value creation.

Thirdly, digital technology enhances the speed and security of service and product delivery, boosting a firm's competitive edge, especially in sectors where digital adoption is minimal, such as physical sciences and

Technovation 129 (2024) 102888

astrophysics, mathematics, and chemical sciences. Other domains like engineering, medical, and healthcare are already familiar with digital technology, implying that while further tech adoption is necessary, the incremental benefits diminish compared to the potential within natural sciences.

Fourthly, digitally advanced spinoffs enable global data sharing, especially using cloud technology. This interconnectivity is vital for labs focused on physics, astrophysics, and chemistry. Compared to other scientific fields, in natural science, simulating materially intense and expensive lab processes can significantly cut down experiment time and foster open-source innovation (Chesbrough et al., 2014), slashing transaction and managerial costs for data gathering, analysis, and simulation (Lu et al., 2022).

Fifthly, digitally advanced spinoffs in natural sciences are better equipped to oversee and integrate data across entities, using tools like cloud technology and live-tracking for lab experiments in fields such as physical sciences, astrophysics, and chemistry. This capability lowers costs and enhances collaboration between major stakeholders. Digitally advanced spinoffs can streamline operations, hasten product development, and communicate research findings more efficiently if they widely adopt digital technologies. This stakeholder engagement is vital for growth and access to external financing, emphasizing the importance of developing digital technologies for intrapreneurial spinoffs.

Finally, implementing digital technologies can boost the speed of experiments and data preparation, influencing product development and validation. Collaborative online engagement has proven efficient, as witnessed during the COVID-19 pandemic (Zhang et al., 2021). Outsourcing to locations where experimentation is financially viable and feasible using digital tech is a trend. Digital tools and collaboration help mitigate the risks inherent in the knowledge commercialization process.

Given the distinctions between "digitally-driven" and "digitally uncertain" spinoffs, it's plausible to assert that digitally-driven intrapreneurial spinoffs in natural sciences will utilize the advantages of digital technologies to maximize the applicability of basic knowledge, including via collaboration with external stakeholders and investors. Their digital capabilities enable rapid material experimentation, process simulations, and knowledge exchange, resulting in more significant knowledge spillovers (Knockaert et al., 2011; Audretsch et al., 2021) and, consequently, increased profits and investment (Rodríguez-Gulías et al., 2018). Investors are likely to anticipate considerably higher returns from digitally-driven spinoffs compared to their uncertain counterparts (Connected Commerce Council, 2021). A case in point is the Kromek spinoff in physics. Led by materials scientist Arnab Basu, this radiation-detection technology developer ventured into countering COVID-19 using digital tech, emphasizing digital marketing and management as vital functions.

However, the digital technologies pioneered by intrapreneurial spinoffs in natural sciences are seldom ready for market at inception. These firms will often need more seed and venture capital for growth (Clarysse et al., 2007; Mariani and Belitski, 2022; Cumming et al., 2021). Therefore, we hypothesize that digitally driven spinoffs will achieve greater profitability and access to external finance. Therefore we hypothesize that spinoffs that are digitally driven will achieve greater profitability and access to external finance.

H2. Digitally-driven intrapreneurial academic spinoffs in the natural sciences increase their access to equity and debt capital to a greater extent than digitally-uncertain spinoffs in natural sciences and other fields of science.

H3. Digitally-driven intrapreneurial academic spinoffs in the natural sciences increase their profits to a greater extent than digitally-uncertain spinoffs in natural sciences and other fields of science.

4. Methodology

4.1. Sample

Most datasets represent a static combination of resources at a particular point in time, as conducting longitudinal surveys for potential dynamic analysis is even more costly and time-intensive. Self-reported data from firms is likely to be more biased (Gonzalez et al., 2012) than objective data, especially if managers in growing firms cannot gather enough data on markets, customers, and competitors or underreport data for accounting reasons.

The data agency Beauhurst collects firm-level data for all high-growth firms in the UK (Beauhurst, 2022). This data provides information (daily, monthly, and annually) on all high-growth firms in the UK, including all university spinoffs from 2011 onwards. Beauhurst employs an artificial intelligence algorithm to extract information from websites and annual reports about existing and potential high-growth firms, including university spinoffs. We obtained a comprehensive sample of university (academic) spin-offs identified by Beauhurst as academic spinoffs and officially registered in the UK. This sample comprised 1194 academic spinoffs, including 1047 academic spinoffs in non-natural science fields. These 1194 academic spinoffs were observed from 2015 to 2019 as a panel. However, not all academic spinoffs are associated with a university or its staff (Klofsten et al., 2021).

To distinguish intrapreneurial academic spinoffs from the broader cohort of academic spinoffs, we drew on previous research on university spinoffs (Mustar et al., 2008; Guerrero et al., 2016, 2021; Klofsten et al., 2021; Valka et al., 2020). We manually examined data from the LinkedIn accounts of a CEO and (co)founder(s) of an academic spinoff (provided within the Beauhurst data). We verified whether a (co) founder(s), a CEO, or both, held full-time research positions (e.g., researcher, assistant, associate, or full professor) at a UK university. University spinoffs with a founder, CEO, or both who had ceased their academic careers and were no longer full-time university employees between 2015 and 2019 (the data collection period) were excluded from our sample. This criterion reduced the number of academic spinoffs in our sample from 147 to 89 in the natural sciences and from 1047 to 660 in non-natural science fields. In cases with conflicting information about employment years or professional roles, we manually checked university websites to determine if a (co)founder or a CEO was listed on the official university/faculty page and whether that information remained current. Consequently, having a full-time professional position at a UK university was a crucial criterion for inclusion in our sample as an intrapreneurial academic spinoff.

Therefore, our final sample consisted of 89 natural science intrapreneurial spinoffs and 660 non-natural science intrapreneurial spinoffs from 2015 to 2019. The number of observations remained consistent across years, allowing us to analyze how independent characteristics influenced the dependent variables consistently. Please refer to Table 1 for sample descriptions of both natural science and non-natural science intrapreneurial spinoffs.

Drawing on OECD (2015) Frascati Manual we describe our sample construction as follows. Intrapreneurial spinoffs in our data which represent natural sciences belong to physical sciences and astrophysics, mathematics, chemical sciences, earth and related environmental sciences. Our sample of natural sciences does not include spinoffs from Computer and information sciences as well as biological sciences.

Intrapreneurial spinoffs in our data which represent non-natural sciences belong to Engineering and Technology (49.35%); Medical and Health Sciences (25.63%); Social Sciences (10.41%); Agricultural Sciences, including agriculture, forestry, fisheries, veterinary medicine, animal and dairy science, veterinary science, agricultural biotechnology (10.12%); and Humanities (4.49%).

Most firms in the sample of natural science academic spinoffs are at the seed stage (start-ups) (40.01% of the sample), the venture stage—early growth (20.02% of the sample), and the growth stage (7.05%).

Table 1

Descriptive summary statistics.

Type of academic	spin-off	Natural s	science $= 8$	9 spinoffs	:	Other fie	elds = 660	spinoffs	
Variables	Description	Mean	St.dev	Min	Max	Mean	St.dev	Min	Max
Grants	Number of grants received	3.84	3.85	0.00	24.00	2.76	3.51	0.00	27.00
Fundraise	Total equity fundraising, in GBP in logarithms	10.63	6.00	0.00	18.48	11.99	6.21	0.00	20.31
Bank loans short term	Total bank short term loans in logarithms	3.42	5.46	0.00	17.42	2.67	5.03	0.00	18.27
Bank loans long term	Total bank long term loans in logarithms	6.68	6.26	0.00	17.42	6.03	6.12	0.00	18.64
Profits	Total profit after tax in logarithms	1.01	3.60	0.00	16.19	0.55	2.74	0.00	18.36
Seed stage	Seed stage $= 1$, zero otherwise	0.34	0.47	0.00	1.00	0.36	0.48	0.00	1.00
Venture stage	Venture stage $= 1$, zero otherwise	0.31	0.47	0.00	1.00	0.35	0.48	0.00	1.00
Growth stage	Growth stage $= 1$, zero otherwise	0.06	0.25	0.00	1.00	0.08	0.27	0.00	1.00
Established stage	Established stage $= 1$, zero otherwise	0.11	0.32	0.00	1.00	0.05	0.21	0.00	1.00
Pinterest	Spin-off has a Pinterest account $= 1$, zero otherwise	0.02	0.19	0.00	1.00	0.07	0.25	0.00	1.00
Instagram	Spin-off has Instagram account $= 1$, zero otherwise	0.20	0.40	0.00	1.00	0.25	0.43	0.00	1.00
Twitter	Spin-off has tweeter account $= 1$, zero otherwise	0.70	0.33	0.00	1.00	0.76	0.43	0.00	1.00
LinkedIn	Spin-off has LinkedIn account $= 1$, zero otherwise	0.82	0.33	0.00	1.00	0.88	0.32	0.00	1.00
Women CEO	Women is Chief executive officer $= 1$, zero otherwise	0.09	0.29	0.00	1.00	0.08	0.28	0.00	1.00
Intangible assets	Intangible assets, GBP in logarithms	4.01	5.72	0.00	15.42	4.33	5.72	0.00	18.35
Working capital	Working capital, GBP in logarithms	9.36	5.36	0.00	1.89	10.46	5.71	0.00	19.05
Fundraising event	Number of fundraising events	3.65	3.23	0.00	12.00	3.36	3.05	0.00	17.00
Digitally-driven	Spinoff uses or develops i) digital technology (platform) aiming to disrupt the existing market is developed by a firm $= 1$, zero otherwise and ii) firm names digital technology in its business model description on the website $= 1$, zero otherwise	0.78	0.41	0.00	1.00	0.25	0.43	0.00	1.00
University age	University age (years), where spinoff is established	311.14	169.69	52.00	500.00	291.90	164.66	26.00	500.00
Firm size	Number of full-time employees in logarithms	2.01	0.97	0.60	4.10	2.41	1.06	0.69	6.42

Source: Academic spin-offs Beauhurst (2022).

Most firms in non-natural science academic spinoffs are at the seed stage (start-ups) (44.21% of the sample), venture stage—early growth (19.01% of the sample), and growth stage (3.05% of the sample). A low share of established or exit-stage firms can be explained by the emerging characteristics of the industries and the newness of the technology used in the industry, with many firms still at the experimental stage.

4.2. Variables

4.2.1. Dependent variable

We have five dependent variables. The first variable is the number of grants received from both public and non-public sponsors. Innovation grants are mainly provided by the UK innovation agency, which is part of the UK Research and Innovation. The second variable is a total equity fundraising in GBP in logarithms, which demonstrates access to equity funding including venture capital (VC) and business angels. The third variable is total bank short-term loans in logarithms, which is a proxy for access to debt finance. Finally, we estimate firm efficiency in value capture and profitable growth by adding total profits in logarithms in our model.

4.2.2. Explanatory variables

We use the following explanatory variables to test our research hypothesis. Firstly, to test H1a and H1b we used a set of binary variables associated with an intrapreneurial spinoff growth stage, such as seed stage, venture stage, growth stage, established stage, exit stage, dead and dormant stages (H1a-H1b) building on pripor research of firm growth stages (Churchill and Lewis, 1983; Belitski and Desai, 2021). Spinoffs that are at exit, dead and dormant stag is our reference category.

We draw on prior research of Giones and Brem (2017) in identifying the digital technology and we apply the Connected Commerce Council (2021) definition to distinguish between intrapreneurs that are digitally-driven (meaning they have developed new digital technologies and adopted tools they have found essential for their business model) and intrapreneurs that are digitally-uncertain (meaning they have not developed and/or adopted digital technology). As an example, new digital technologies could be developed as part of the Internet of Things, Industry 4.0, artificial intelligence, block chain, automation, remote monitoring, predictive maintenance, smart contracts, big data, and cloud computing (Bock et al., 2018).

In order to test our H2 and H3 related to the relationship between digitally-driven spinoffs and access to finance (H2) and profitability (H3), we used a variable called "digitally driven" which is a binary variable equals one if a firm developed digital technology in-house which is market disruptive, zero otherwise. In addition, we controlled whether a spinoff reported a use of digital technology in the description of their business model or not. In case of no evidence and mentioning using the digital technology is not essential for their business– we considered that digital technology is not essential for their business and hence a firm would be changed for digitally-uncertain. The description of the digital technology on the spinoff website could include development and adoption of new technology, development of technology which is used for product development and testing, transfer and engagement and developed by the company or outsourced to the third parties (Aspelund et al., 2005).

4.2.3. Control variables

We used the number of successful fundraising events as a control for financial investment readiness. We use the binary variable "women-led firm" (1 if a CEO is female, 0 if not) to measure the performance of women-led firms. Female CEOs might have different access to finance in the early stages of business growth compared to male CEOs (Alsos and Ljunggren, 2017). Prior research has shown that women are generally more risk-averse, which could influence the type of financing they choose to access (Audretsch et al., 2022a). The heightened risk aversion of female CEOs, combined with operating in a traditionally "masculine world", might limit their access to equity finance. There is recent evidence supporting this in the context of female leadership in academic spinoffs (Lauto et al., 2022), which could lead to reduced innovation and profits (Audretsch et al., 2022b).

We control for firm size, which is measured as the log of total assets, to account for potential diminishing marginal returns as firm size increases.

Table 2

Results of the fixed effects regression: Dependent variables - access to finance and firm performance.

Specification	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Dependent variable	Grants	Grants	Fundraise	Fundraise	Bank loans short term	Bank loans short term	Bank loans long term	Bank loans long term	Profits	Profits
Type of spin-off	Natural sciences	Other fields	Natural sciences	Other fields	Natural sciences	Other fields	Natural sciences	Other fields	Natural sciences	Other fields
Seed stage (H1)	-3.27**	-0.02	-0.39	-0.40	-4.66***	-3.02***	-4.99***	-2.58***	-1.73*	-1.68***
17	(1.60)	(0.33)	(1.80)	(0.58)	(1.65) -4.10***	(0.70)	(1.76)	(0.78)	(0.84)	(0.47) 1.69***
Venture stage	-4.72**	1.18***	1.37 (1.44)	0.05 (0.14)		2.09** (0.90)	-2.85 (1.80)	1.84** (0.76)	-3.53*	
(H1)	(1.80)	(0.40)	1 00 (1 0 0	1 00++	(1.52)	1 00++ (0 01)	0.51 (0.04)	0.00 (0.00)	(1.87)	(0.50)
Growth stage	4.08 (2.63)	1.87**	1.29 (1.04)	1.38**	2.25***	1.89** (0.91)	3.71 (2.34)	0.29 (0.96)	4.53**	1.97***
(H1)		(0.90)		(0.60)	(0.72)				(2.25)	(0.60)
Established	2.74 (1.90)	0.43 (1.06)	4.13**	2.18***	4.66***	0.45** (0.24)	4.90** (2.25)	0.67 (1.22)	-1.68	-0.63
stage (H1)			(2.03)	(0.51)	(1.72)				(2.36)	(0.95)
Pinterest	-4.86***	-0.16	-1.85	-1.46**	-3.21 (2.11)	-0.08 (0.78)	-1.93 (3.74)	-0.98 (0.91)	1.72	-0.22
	(1.44)	(0.51)	(2.68)	(0.68)					(3.73)	(0.36)
Instagram	-1.69*	-0.41	0.65 (1.16)	-0.34	-0.73 (1.40)	0.05 (0.45)	1.78 (1.67)	0.09 (0.57)	-1.20	-0.03
	(0.94)	(0.35)		(0.40)					(0.77)	(0.26)
Twitter	1.48*	0.54*	1.08**	0.92*	1.77** (0.70)	0.15** (0.06)	1.29** (0.60)	0.47 (0.28)	1.25	0.20 (0.11)
	(0.79)	(0.31)	(0.50)	(0.50)					(0.70)	
LinkedIn	1.09 (1.00)	0.28 (0.37)	2.86*	1.76***	2.03 (1.70)	0.01 (0.58)	-1.13 (2.15)	0.64 (0.69)	1.66*	0.09 (0.28)
			(1.47)	(0.66)					(0.91)	
Women CEO	1.09 (1.52)	0.07 (0.39)	0.82 (1.41)	-0.07 (0.62)	-0.89 (1.16)	1.15 (0.70)	0.19 (2.58)	1.15 (0.81)	-2.12 (1.45)	0.22 (0.35)
Intangible	0.05***	-0.01	0.05***	0.100***	0.05***	0.06** (0.03)	0.07** (0.03)	0.07* (0.04)	0.09**	0.01**
assets	(0.01)	(0.03)	(0.01)	(0.03)	(0.01)	. ,			(0.03)	(0.00)
Working	0.03 (0.07)	-0.03	0.07***	0.09***	-0.48***	-0.27***	-0.41***	-0.19***	0.02**	0.03*
capital		(0.03)	(0.2)	(0.03)	(0.09)	(0.04)	(0.11)	(0.04)	(0.00)	(0.02)
Fundraising	0.22*	-0.05	1.30***	1.16***	-0.62***	-0.06 (0.08)	-0.30**	-0.10 (0.09)	-0.01	-0.09
event	(0.13)	(0.05)	(0.21)	(0.09)	(0.17)		(0.15)		(0.21)	(0.07)
Digitally-	0.09 (0.61)	0.88**	0.91***	0.34***	1.45***	-0.14 (0.43)	1.96** (0.68)	-0.48 (0.52)	1.19**	0.45**
driven (H2)		(0.35)	(0.25)	(0.12)	(0.27)				(0.46)	(0.18)
University age	0.01***	-0.01***	0.01***	0.01***	0.01***	0.01** (0.00)	0.01 (0.00)	-0.01 (0.00)	-0.01	0.01 (0.00)
oniversity age	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	0101 (0100)	0.01 (0.00)	(0.00)	0.01 (0.00)
Firm size	0.53 (0.56)	0.80***	-0.42	0.22 (0.22)	1.47** (0.70)	1.12***	1.90** (0.84)	1.75***	0.69	0.45**
1 1111 5120	0.00 (0.00)	(0.22)	(0.63)	0.22 (0.22)	1.17 (0.70)	(0.27)	1.90 (0.01)	(0.28)	(0.49)	(0.20)
Constant	2.12 (1.83)	0.62 (0.48)	2.49 (2.53)	2.88***	6.03** (2.40)	4.99***	8.32***	6.05***	1.26	0.56 (0.46)
Constant	2.12 (1.05)	0.02 (0.40)	2.49 (2.33)	(0.80)	0.05 (2.40)	(0.90)	(3.01)	(1.00)	(2.37)	0.50 (0.40)
Number of spinoffs	89	660	89	660	89	660	89	660	89	660
Number of obs.	356	2640	356	2640	356	2640	356	2640	356	2640
R2			.692	.523			.407	.386		.317
	.411	.330			.546	.304			.287	
F-statistics	6.70	5.55	13.63	33.81	11.74	7.49	6.92	11.42	5.68	6.82
Log-likelihood	-197.12	-1717.52	-211.66	-1896.97	-214.45	-1927.35	-235.89	-2063.06	-199.33	-1559.10

Note: Standard errors in parentheses. Significance levels: *** 1%, ** 5%, * 10%; Industry and regional fixed effects are oppressed to save space. Reference region = Northern Ireland; reference industry – finance and financial technologies. Standard errors robust for heteroskedasticity are in parenthesis. Source: Academic spin-offs Beauhurst (2022).

Measuring firm size by its total assets and working capital is particularly fitting for firms in emerging industries. This is because many of these firms undergo rapid growth and scaling (Belitski et al., 2023) in the years following their product introductions. We consider the age of the university where the firm was established, as it might correlate with potential access to resources for intrapreneurial spinoffs (Heblich and Slavtchev, 2014; Audretsch and Belitski, 2021b, 2022). We assess the extent of digital presence and the role of social networks, which could be crucial for fundraising, visibility, and sales. To do this, we use a set of binary variables indicating whether businesses are active on platforms like LinkedIn, Pinterest, Twitter, and Instagram. We've incorporated 12 regional fixed effects for the UK as identified by the Office of National Statistics. This encompasses nine regions in England (South-East of England, South-West of England, London, East of England, West Midlands, East Midlands, North-East, North-West, and Yorkshire and Humber) and three countries: Wales, Scotland, and Northern Ireland. The reference category is Northern Ireland. All control variables are considered with a one-year lag.

4.3. Estimation strategy

We employ a random-effects panel data estimation with industry and regional fixed effects. Our dependent variables are continuous, and the data is panel of four years (Wooldridge, 2009). Our dependent variable is y_{it} (firm's ability to raise funding – grant numbers, equity funding, short-term

$$y_{it} = \beta_0 + \beta_1 S_{it-1} + \beta_2 D_{it-1} + \beta_3 x_{it-1} + \delta_z + \omega_r + u_{it}$$
(1)

Vector S_{it-1} is a set of explanatory variables related to the growth stage of intrapreneurial spinoff i at time t – 1; vector D_{it-1} is a binary variable equals one if an intrapreneurial spinoff i at time t – 1 is digitallydriven, zero otherwise; x_{it-1} is a vector of control variables of the intrapreneurial spinoff i at time t, and u_{it} is an error term (Wooldridge, 2009). Vectors δ_x , ω_r are industry - (z) and region (r) fixed effects. A bootstrapping of errors was also applied, which led to similar results in terms of the sign and significance of all confidants but of a different size.

We started by exploring the multicollinearity of the variables by examining the variance inflation factors for all variables, finding each less than 5. In addition, we analyzed the correlation coefficients, ensuring that no coefficients were greater than 0.70. We analyzed all the variables' histograms and found the errors were identically and independently distributed with constant variance.

5. Results

Table 2 presents the regression results testing our H1 and H2. Our H1a posits that intrapreneurial academic spin-offs in natural sciences raise equity and debt capital at later stages of firm growth compared to intrapreneurial academic spin-offs in other science fields. This hypothesis is supported.

Intrapreneurial spinoffs in natural sciences are less likely to secure grant funding during the seed stage ($\beta = -3.27$, p < 0.01) and venture stage ($\beta = -4.72$, p < 0.01) compared to the exit stage and compared to spinoffs in other science fields at the seed stage ($\beta = -0.02$, p > 0.10) and venture stage ($\beta = 1.18$, p < 0.01) (spec. 1, Table 2). Intrapreneurial spinoffs in other science disciplines continue to show higher grant fundraising during the growth stage ($\beta = -0.02$, p > 0.10) and venture stage ($\beta = 1.18$, p < 0.01), in comparison to the exit stage (spec. 2, Table 2).

Intrapreneurial spinoffs in natural sciences are less likely to raise equity funding during the growth stage than during the exit stage and compared to intrapreneurial spinoffs in other science fields ($\beta = 1.38$, p < 0.05) (spec. 4, Table 2). Notably, intrapreneurial spinoffs in natural sciences at the established stage ($\beta = 4.13$, p < 0.001) (spec. 3, Table 2) almost double their equity funds when compared to intrapreneurial spinoffs in other fields ($\beta = 2.18$, p < 0.001) (spec. 4, Table 2).

Both intrapreneurial spinoffs in natural sciences and other fields secure fewer short-term loans during the seed stage than at the exit stage. Yet, intrapreneurial spinoffs in natural sciences continue to attract lower short-term bank capital during the venture stage ($\beta = -4.10$, p < 0.05) (spec. 5, Table 2), while intrapreneurial spinoffs in other fields obtain more debt financing in comparison to the exit stage ($\beta = 2.10$, p < 0.01) (spec. 6, Table 2). Both intrapreneurial spinoffs in natural sciences and other disciplines have fewer long-term bank loans at the seed stage ($\beta = -4.99$, p < 0.01) and ($\beta = -2.58$, p < 0.01) respectively (spec. 7–8, Table 2). While intrapreneurial spinoffs in other fields can access more long-term bank capital during the venture period ($\beta = -1.84$, p < 0.01), those in natural sciences do not. However, the increase in bank lending for spinoffs in natural sciences is significant at the established stage ($\beta = 4.90$, p < 0.01) compared to the exit stage, and it's considerably higher than intrapreneurial spinoffs in other fields ($\beta = 0.67$, p > 0.10) (spec. 7–8, Table 2).

Venture capitalists seem to have a bias related to academic spin-offs at the early stages of firm growth, as these investors struggle to identify and evaluate the technologies used by intrapreneurs (Munari and Toschi, 2010). While we have previously highlighted that radically new technologies can be complicated and difficult to evaluate, increasing the magnitude of information asymmetries and uncertainty, they also have a great potential to generate disruptive technologies.

Our H1b, which posits that intrapreneurial academic spin-offs in natural sciences increase their profits at later stages of firm growth compared to intrapreneurial academic spin-offs in other science fields, is partly supported. The regression coefficient for natural science intrapreneurial spin-offs remains negative and statistically significant at the seed stage ($\beta = -1.73$, p < 0.05) and venture stage ($\beta = -3.53$, p < 0.05) (spec. 9, Table 2). In contrast, profitability of spin-offs in other sciences increases at the seed stage compared to the exit stage ($\beta = 1.69$, p < 0.001) (spec. 10, Table 2). In the growth stage, intrapreneurial spin-offs in natural sciences, compared to the exit stage and spin-offs in non-natural sciences, demonstrate higher profits ($\beta = 4.53$, p < 0.01) (spec. 9, Table 2). Interestingly, profit growth at the established stage is consistent with the exit stage for both types of intrapreneurial spin-offs.

Our H2a, which posits that digitally-driven intrapreneurial academic spin-offs in the natural sciences raise more equity and debt capital than digitally-uncertain firms in the natural sciences and spin-offs in other science fields, is supported. Digitally-driven intrapreneurial academic spin-offs in other fields receive, on average, a higher number of grants ($\beta = 0.88$, p < 0.01) (spec. 2, Table 2) compared to digitally-driven spin-offs in natural science ($\beta = 0.09$, p > 0.10) (spec. 1, Table 2). We find that the adoption of digital technology enables intrapreneurial academic spin-offs in natural sciences to raise more equity funding ($\beta = 0.91$, p < 0.001) (spec. 3, Table 2) than other spin-offs ($\beta=0.34, p<0.001$) (spec. 4, Table 2). Notably, for both short-term and long-term bank finance, we find that digitally-driven intrapreneurial spin-offs in natural science can secure more short-term bank finance ($\beta = 1.45$, p < 0.01) (spec. 5, Table 2) and long-term bank finance (β = 1.96, p < 0.01) (spec. 7, Table 2) compared to spin-offs in other fields. This furthers prior research on the role of digitalization for small and mediumsized enterprises (Connected Commerce Council, 2020, 2021) and in

university spin-offs (Shane, 2004; Rippa and Secundo, 2019). The coefficient for digitally-driven intrapreneurial startups in other sciences is negative and not statistically significant. This suggests that intrapreneurial spin-offs in natural science are more likely to obtain bank finance if they signal to lenders as digitally-driven intrapreneurial spin-offs. We also find that digitally-driven intrapreneurial spin-offs in natural sciences have a higher level of equity than debt finance, which underscores the role of VCs in supporting university spin-offs (Rodríguez-Gulías et al., 2018; Bock et al., 2018).

Our H3, which posits that digitally-driven intrapreneurial academic spin-offs in the natural sciences achieve greater profits than digitallyuncertain firms in the natural sciences and spin-offs in other science fields, is supported. Intrapreneurial spin-offs in natural sciences that adopt digital technology report, on average, a 1.19 percent ($\beta = 1.19$, p < 0.01) (spec. 9, Table 2) higher profit than natural science spin-offs that aren't digitally driven. The coefficient is also higher compared to the returns on a digitally-driven approach for academic spin-offs in other sciences ($\beta =$ 0.45, p < 0.01) (spec. 10, Table 2). This indicates that returns on investment in digital technology and its adoption in business models for profitability are significantly higher for spin-offs in natural sciences compared to spin-offs in other fields. Intrapreneurial academic spin-offs see a greater increase in sales and profits when digital technologies enhance collaboration with external stakeholders and facilitate knowledge exchange using digital tools, thereby accelerating knowledge creation and transfer (Cassiman and Veugelers, 2002; Audretsch and Belitski, 2020).

6. Discussion

Intrapreneurial academic spin-offs serve as crucial conduits in the commercialization of university technologies and knowledge (Pinchot, 1985; Guerrero et al., 2016; Klofsten et al., 2019). University researchers are more inclined to engage in intrapreneurial activities and exhibit intrapreneurial behavior across various fields when their personal expertise and intrapreneurial capabilities receive support from the university in the form of a spin-off (Klofsten et al., 2021).

Like any new venture, intrapreneurial academic spin-offs face financial constraints and aim to source external resources to bridge the funding gap (Mustar et al., 2008; Rasmussen and Sørheim, 2012; Sørheim et al., 2011). This study investigates the differences in access to grants, equity, and debt finance, as well as firm profitability between intrapreneurial academic spin-offs in natural sciences and those in other fields. Using micro-level data from 89 intrapreneurial spin-offs in natural sciences and 660 in other fields from the UK (2015-2019), we analyzed their financial access and profitability across different growth stages. Our findings suggest that early-stage intrapreneurial spin-offs in natural science are perceived as riskier, with more uncertain investments than those in other science fields. Consequently, natural science spin-offs secure fewer public grants and raise less debt finance on average. Despite their innovative nature, the innovations often demand extra investment and validation. This prolonged period from invention to market introduction limits fundraising during the initial stages (Sørheim et al., 2011; Bock et al., 2018).

Our empirical results confirm that intrapreneurial spin-offs in natural sciences secure funding at later growth stages compared to their counterparts. Digitally-driven academic spin-offs in natural science possess a pronounced advantage in fundraising and accessing both short and long-term bank loans over both intrapreneurial spin-offs in other sciences and digitallyuncertain spin-offs in natural sciences. Furthermore, they also achieve higher profits than digitally-driven spin-offs in other fields, reinforcing previous research (Giones and Brem, 2017; Secundo et al., 2020). These findings highlight that intrapreneurial spin-offs from natural sciences benefit more from the adoption and in-house development of digital technologies compared to those from other fields. Emphasizing in-house technology development proves more conducive for profitability than returns on digital tech development in alternative science fields. Investments in internal R&D and human capital to become digitally-driven are crucial. Being digitally-driven can significantly reduce operational, logistics, and transaction costs within spin-offs and enhance collaborations with external

stakeholders (Björkdahl, 2020), thereby amplifying information exchange.

Our results demonstrated that the substantial returns on digitally-driven technologies in natural science spin-offs may primarily relate to signaling to stakeholders like investors, governments, lenders, and customers (Giones and Brem, 2017; Cumming et al., 2021). For example, a spin-off may be better equipped to engage with external partners, showcase its innovations, and participate in open innovation strategies (Chesbrough et al., 2014). Investors likely view digitally-driven intrapreneurial spin-offs, especially in natural science, favorably. Many intrapreneurial academic spinoffs struggle to secure high profits and access to grants, equity, and debt financing during their initial growth stages. This can be attributed to challenges like technological risks, market uncertainties (Clarysse et al., 2007), a digital divide (Audretsch and Belitski, 2021a), and a lack of intrapreneurial skills (Guerrero et al., 2015, 2016). We found that digitally advance spinoffs achieve double the equity funding and greater access to bank capital than their digitally uncertain spinoffs.

Our study's results extend previous research on signaling for external finance access by intrapreneurial spin-offs (Conti et al., 2013; Giones and Brem, 2017; Secundo et al., 2020). Consistent with Conti et al. (2013), we found that the amount founders invest positively correlates with business investment.

Drawing upon the existing literature on universities' intrapreneurial capabilities (Meoli and Vismara, 2016; Klofsten et al., 2019, 2021), we argue that it is imperative for universities to integrate intrapreneurship into their business models. Such integration would entail a robust commitment from university managers and researchers (Guerrero and Urbano, 2014; Audretsch and Belitski, 2021b). The objective is to foster a conducive environment for intrapreneurial behavior, supported by clear leadership, defined university policies, systematic incentives, and transparent communication procedures (Klofsten et al., 2019).

Although the current body of research provides valuable insights into academic entrepreneurship in both developed (Klofsten and Jones-Evans, 2000; Guerrero and Peña-Legazkue, 2013) and developing countries (Belitski et al., 2019), there is still a dearth of knowledge about how universities balance research, teaching, and commercialization—their three primary missions (Guerrero et al., 2020, 2021). Moreover, the success of raising funds for intrapreneurial spinouts is not merely contingent on the surrounding institutions, be they formal or informal (Guerrero and Urbano, 2012, 2014), or on the entrepreneurial ecosystems of universities (Belitski and Heron, 2017; Guerrero et al., 2020). Instead, it is intrinsically tied to the intrinsic capabilities of the intrapreneurial spinoffs themselves (Guerrero et al., 2015; Klofsten et al., 2021).

Historical research has expanded our comprehension of entrepreneurial universities as entities (Guerrero et al., 2016; Radko et al., 2022). Yet, there remains a gap in our understanding of the intrapreneurial mechanisms within these institutions, as emphasized by Klofsten et al. (2019).

We address this gap by highlighting the nuanced needs of intrapreneurial academic spinoffs based on their scientific domain. These spinoffs, particularly those in natural sciences, often face longer paths to commercialization, requiring persistent support and investment from the university. Due to the nature of their products and the expertise they necessitate, these spinoffs frequently require extended timelines to develop products, secure funding, and achieve commercialization. We postulate that the utilization of digital technologies can expedite these processes, especially when coupled with simulations for experimentation and open innovation strategies (Li et al., 2016; Lu et al., 2022).

Our study highlights the importance of investment in both digital and intrapreneurial capabilities, enabling spinoffs to adapt swiftly to changes and craft innovation pathways conducive to funding and profitability (Wright et al., 2006; Belitski et al., 2023).

7. Conclusions

Building upon previous research on intrapreneurial activities in universities (Bienkowska and Klofsten, 2012; Audretsch and Belitski, 2013; Bienkowska et al., 2016; Meoli et al., 2019), this study theorizes and empirically tests several boundary conditions for knowldge transfer in intrapreneurial spinoffs in the UK, particularly concerning their access to finance and profitability. This includes factors such as the growth stage of the spinoff (Cunningham et al., 2022), the field of science, and the development and utilization of digital technologies (Clarysse et al., 2005, 2007; Rippa and Secundo, 2019). We specifically spotlight digitally driven intrapreneurial spinoffs, viewing them as catalysts for economic efficiency and tools to signal external stakeholders about a spinoff's quality and readiness for the digital era.

Using micro-level data from 2015 to 2019 on 89 natural science intrapreneurial academic spinoffs and 660 from other scientific domains in the UK, we discern significant takeaways for university leaders and policymakers. For instance, spinoffs in natural sciences like physics and chemistry often differ in their financing strategies and timelines compared to those in other academic disciplines. The intricate nature of spinoffs emerging from natural science-focused institutions is often linked to the integration of intrapreneurial and digital capabilities central to their business models, enabling collaboration with spinoffs from diverse scientific backgrounds. Additionally, these spinoff managers must understand the intricacies of their technologies, necessitating a prolonged journey from idea creation to commercialization. For such spinoffs, internally developed digital technologies can be pivotal in fostering relationships with investors, policymakers, and other stakeholders. The in-house progression of such technologies accelerates their transition from tangible labs to virtual simulations.

Furthermore, we advocate for intrapreneurial spinoff managers to invest in digital technologies. This will provide better market oversight and foster global collaborations, promoting open innovation and the Triple-Helix model (Brem and Radziwon, 2017; Jovanovic et al., 2021; Audretsch et al., 2022c) and will significantly increase access to equity and debt finance at the earlier stages of growth.

Our study is not without limitations, which pave the way for future research. Our first limitation concerns the relatively low number of natural science academic spinoffs in the UK. With a count of 147, which further dwindles to 89 after considering specific affiliations, our analysis options regarding financial access and profitability were constrained. Rather than a collective analysis, we separately assessed spinoffs in natural sciences and other disciplines, gauging the impact magnitude for each spinoff type. Subsequent research might employ varied methodologies, such as case studies and interviews, and examine the replicability of our findings beyond UK-centric academic spinoffs. Our second limitation arises from our inability to differentiate between spinoffs where leadership has shifted away from the university, despite the institution retaining control. Although instances of this might be limited, such spinoffs were excluded from our study. Future investigations might delve deeper into the ownership dynamics of these spinoffs.

Prospective studies can also explore other boundary conditions impacting financial access and profitability, such as human and technological resource constraints, geographical influences, and the significance of entrepreneurial university ecosystems researching how we can gauge the financial preparedness of intrapreneurial spinoffs for diverse finance. Moreover, the role of Technology Transfer Offices (TTOs) and commercialization units in assisting academic spinoffs, rather than adhering to conventional knowledge licensing models, warrants further exploration. The crucial role of TTO support for digital intrapreneurs is well-established (Meoli and Vismara, 2016); however, tailoring this assistance based on the scientific domain of the spinoff remains a compelling avenue (Heirman and Clarysse, 2004; Abreu and Grinevich, 2013).

Data availability

The authors do not have permission to share data.

Appendix. A

Table A1 Correlation matrix for spinoffs in natural sciences

$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	1	2	3	4	5	9	7	8	6	10	11	12	13	14	15	16	17	18	19
	1. Grants 1																		
	0	23* 1																	
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	_		1 *1																
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	-	U	0	1															
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	-			-0.07*	1														
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$				-0.26^{*}	-0.11*	1													
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	Ige			-0.07*	-0.19*	-0.47*	1												
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$				0.32^{*}	0.07*	-0.18	-0.17	1											
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					0.05^{*}	-0.2511^{*}	-0.24^{*}	•0.09	1										
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$					0.20	-0.13	-0.13	-0.05	0.13	1									
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					-0.14	0.04	0.06	0.12	-0.08	0.06	1								
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					0.15	0.23^{*}	0.06	0.14	0.05*	0.11	0.28^{*}	1							
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					0.10	0.21	0.25^{*}	0.09	-0.22*	0.07	0.19	0.32^{*}	1						
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					-0.08	0.06	-0.11	-0.08	0.16^{*}	0.17	0.06	0.07	0.11	1					
$ \begin{array}{llllllllllllllllllllllllllllllllllll$					-0.08	-0.05	0.12	0.05^{*}	-0.06	-0.14	-0.05	0.16	0.01	-0.21*	1				
nt 0.27^{*} 0.61^{*} -0.07^{*} 0.18 0.01 -0.47^{*} 0.26^{*} 0.43^{*} -0.10 0.06 0.06 0.25^{*} 0.25^{*} 0.06 1 0.05^{*} 0.11^{*} 0.14^{*} 0.16^{*} 0.05 0.03 -0.10 0.13^{*} -0.10 0.10^{*} 0.11 0.06 0.17 0.05 0.09^{*} 0.17 0.13^{*} 0.11^{*} 0.08 0.06 -0.02 -0.09 0.03 -0.07^{*} -0.03 0.04 0.07 -0.04 0.37^{*} 0.26^{*} 0.36^{*} 0.17^{*} -0.55^{*} 0.13 0.36^{*} 0.07^{*} 0.12 -0.04 0.34^{*} 0.17 -0.15					0.04^{*}	-0.14	0.2124	0.08^{*}	0.04^{*}	-0.05	-0.04	0.06	0.02	0.01	0.02^{*}	1			
$\begin{array}{cccccccccccccccccccccccccccccccccccc$					0.01	-0.47*	0.26^{*}	0.43^{*}	-0.10	0.06	0.06	0.25^{*}	0.25^{*}	0.06	0.17^{*}	0.16	1		
$\begin{array}{cccccccccccccccccccccccccccccccccccc$				0.16^{*}	0.05	0.03	-0.10	0.13^{*}	-0.10	0.10^{*}	0.11	0.06	0.17	0.05	0.13^{*}	0.14^{*}	0.03	1	
0.37* $0.37*$ $0.26*$ $0.36*$ $0.17*$ $-0.55*$ 0.13 $0.36*$ $0.07*$ 0.12 -0.04 $0.34*$ 0.17 -0.15	-			0.11^{*}	0.08	0.06	-0.02	-0.09	0.03	-0.07^{*}	-0.03	0.04	0.07	-0.04	-0.02	-0.01	-0.02	0.10	1
	-			0.36^{*}	0.17^{*}	-0.55^{*}	0.13	0.36^{*}	0.07*	0.12	-0.04	0.34^{*}	0.17	-0.15	0.22^{*}	0.27^{*}	0.51^{*}	0.13	0.18^{*}

Note: Significance level: ** 5% Source: Academic spin-offs Beauhurst (2022).

Table A2

	-	c	¢	•	ц	7	Ľ	0	c	10	11	10	10	77	1	16	17	10	10
	-	1	c	+	n	D		0	٦	TO	1	77	C1	14	CT	01	11/	10	ТĄ
1. Grants	1																		
2.Fundraise	*60.0	1																	
3.Bank loans short term	0.03*	0.05	1																
4.Bank loans long term	0.10^{*}	0.03	0.57*	1															
5.Profits	0.02	0.01	0.21^{*}	0.14^{*}	1														
6.Seed stage	-0.21^{*}	-0.28^{*}	-0.21^{*}	-0.22*	-0.13*	1													
7.Venture stage	0.13^{*}	0.24^{*}	0.04^{*}	0.04^{*}	-0.08	-0.54^{*}	1												
8. Growth stage	0.15^{*}	0.10^{*}	0.07*	0.14^{*}	0.07*	-0.21^{*}	-0.21*	1											
9.Established stage	0.04	0.09*	0.13^{*}	0.13^{*}	0.10^{*}	-0.16^{*}	-0.16^{*}	-0.06	1										
10.Pinterest	0.01	0.01	0.01	-0.01	-0.02	-0.04	-0.01	0.11^{*}	-0.03	1									
11.Instagram	0.01	0.09*	0.02	0.03	-0.01	-0.03	0.04	0.10^{*}	0.01	0.25^{*}	1								
12.Twitter	0.11^{*}	0.19^{*}	0.01	0.03	0.04	-0.07	0.10^{*}	0.08*	-0.01	0.12^{*}	0.29^{*}	1							
13.LinkedIn	0.11^{*}	0.25^{*}	0.01	0.06	0.05	-0.11^{*}	0.13^{*}	0.03	-0.03	0.06	0.14^{*}	0.38^{*}	1						
14.Women CEO	0.05	0.02	0.04	0.02	-0.03	0.02	0.06	-0.02	0.01	0.02	0.05	0.09*	0.06	1					
15.Intangible assets	0.05	0.24^{*}	0.13^{*}	0.16^{*}	0.04	-0.17*	0.04	0.10^{*}	0.04^{*}	0.03	0.04	0.05^{*}	0.06	-0.04	1				
16.Working capital	0.03	0.22^{*}	-0.22^{*}	-0.08*	0.12^{*}	-0.13^{*}	0.09^{*}	0.08^{*}	0.06^{*}	0.01	0.01	0.09*	0.17^{*}	-0.04	0.07*	1			
17.Fundraising event	0.11^{*}	0.60^{*}	0.07	0.10^{*}	-0.06	-0.37^{*}	0.24^{*}	0.20^{*}	-0.01	0.09^{*}	0.16^{*}	0.15^{*}	0.19^{*}	0.04	0.23^{*}	0.12^{*}	1		
18.Digitally driven	0.12^{*}	0.07^{*}	0.02	-0.01	-0.01	-0.11^{*}	0.04^{*}	0.05^{*}	0.09*	-0.03	-0.08*	-0.12*	-0.02	0.01	0.03*	0.02	0.08^{*}	1	
19. University age	-0.07*	0.22^{*}	-0.05	-0.07	0.01	-0.05	0.09^{*}	0.01	-0.01	0.01	0.02	-0.01	0.01	0.07	0.01^{*}	0.13^{*}	0.09*	0.05	1
20 Firm size	0.27*	0.34^{*}	0.22^{*}	0.32*	0.21*	-0.44	0.06	0 28*	0.00*	0.07	014*	0.23*	0 26*	-0.04	0.28*	0 31 *	×07 0	500	0.0554

Technovation 129 (2024) 102888

References

- Abreu, M., Grinevich, V., 2013. The nature of academic entrepreneurship in the U.K.: widening the focus on entrepreneurial activities. Res. Pol. 42 (2), 408–422.
- Alsos, G.A., Ljunggren, E., 2017. The role of gender in entrepreneur-investor relationships: a signaling theory approach. Entrep. Theory Pract. 41 (4), 567–590.
- Antoncic, B., Hisrich, R.D., 2003. Clarifying the intrapreneurship concept. J. Small Bus. Enterprise Dev. 10 (1), 7–24. Aspelund, A., Berg-Utby, T., Skjevdal, R., 2005. Initial resources' influence on new
- venture survival: a longitudinal study of new technology-based firms. Technovation 25 (11), 1337–1347.
- Åstebro, T., Braunerhjelm, P., Broström, A., 2013. Does academic entrepreneurship pay? Ind. Corp. Change 22 (1), 281–311.
- Åstebro, T., Braguinsky, S., Braunerhjelm, P., Broström, A., 2019. Academic entrepreneurship: the bayh-dole act versus the professor's privilege. ILR Review 72 (5), 1094–1122.
- Audretsch, D.B., 2014. From the entrepreneurial university to the university for the entrepreneurial society. J. Technol. Tran. 39, 313–321. Audretsch, D.B., Belitski, M., 2013. The missing pillar: the creativity theory of knowledge
- spillover entrepreneurship. Small Bus. Eco., 41, 819–836. Audretsch, D.B., Belitski, M., 2019. Science parks and business incubation in the United
- Kingdom: evidence from university spin-offs and staff start-ups. In: Science and Technology Parks and Regional Economic Development, Palgrave Macmillan, Cham. pp. 99–122.
- Audretsch, D.B., Belitski, M., 2020. The limits to collaboration across four of the most innovative UK industries. Br. J. Manag. 31 (4), 830-855.
- Audretsch, D.B., Belitski, M., 2021a. Knowledge complexity and firm performance: evidence from the European SMEs. J. Knowl. Manag. 25 (4), 693-713.
- Audretsch, D.B., Belitski, M., 2021b. Three-ring entrepreneurial university: in search of a
- new business model. Stud. High Educ. 46 (5), 977–987. Audretsch, D.B., Belitski, M., 2022. A strategic alignment framework for the entrepreneurial university. Ind. Innovat. 29 (2), 285-309.
- Audretsch, B.D., Belitski, M., 2023. The limits to open innovation and its impact on innovation performance. Technovation 119, 102519.
- Audretsch, D.B., Belitski, M., Caiazza, R., 2021. Start-ups, innovation and knowledge spillovers. J. Technol. Tran. 46 (6), 1995-2016.
- Audretsch, D.B., Belitski, M., Chowdhury, F., Desai, S., 2022a. CEO gender, institutional context and firm exports. Int. Bus. Rev. 31 (5), 102008.
- Audretsch, D.B., Belitski, M., Brush, C., 2022b. Innovation in women-led firms: an empirical analysis. Econ. Innovat. N. Technol. 31 (1-2), 90-110.
- Audretsch, D.B., Belitski, M., Guerrero, M., Siegel, D.S., 2022c. Assessing the impact of the UK's Research Excellence Framework on the relationship between university scholarly output and education and regional economic growth. Acad. Manag. Learn. Educ. 21 (3), 394–421.
- Audretsch, D., Belitski, M., Brush, C., 2023. How to secure an innovation grant for firms in new industries? Gender and resource perspectives. Int. J. Entrepreneurial Behav. Res. https://doi.org/10.1108/IJEBR-02-2022-0183.
- Baruah, B., Ward, A., 2015. Metamorphosis of intrapreneurship as an effective organizational strategy. Int. Enterpren. Manag. J. 11, 811-822.
- Belitski, M., Heron, K., 2017. Expanding entrepreneurship education ecosystems. J. Manag. Dev. 36 (2), 163-177.
- Belitski, M., Aginskaja, A., Marozau, R., 2019. Commercializing university research in transition economies: technology transfer offices or direct industrial funding? Res. Pol. 48 (3), 601-615.
- Belitski, M., Desai, S., 2021. Female ownership, firm age and firm growth: a study of South Asian firms. Asia Pac. J. Manag. 38 (3), 825-855.
- Belitski, M., Boreiko, D., 2022. Success factors of initial coin offerings. J. Technol. Tran. 47 (6), 1690-1706.
- Belitski, M., Martin, J., Stettler, T., Wales, W., 2023. Organizational scaling: the role of knowledge spillovers in driving multinational enterprise persistent rapid growth. J. World Bus. 58 (5), 101461.
- Belz, A., Graddy-Reed, A., Zapatero, F., 2021. Impact of Government Grants on Venture Capital Funding of Deep Technology University Spinoffs. Available at: SSRN 3493973.
- Beauhurst, 2022. Discover the UK most fast growing companies. Available at: https:// www.beauhurst.com/. (Accessed 19 September 2022).
- Bienkowska, D., Klofsten, M., 2012. Creating entrepreneurial networks: academic entrepreneurship, mobility, and collaboration during PhD education. High Educ. 64 (2), 207-222.
- Bienkowska, D., Klofsten, M., Rasmussen, E., 2016. PhD students in the entrepreneurial university-perceived support for academic entrepreneurship. Eur. J. Educ. 51 (1), 56-72.
- Björkdahl, J., 2020. Strategies for digitalization in manufacturing firms. Calif. Manag. Rev. 62 (4), 17–36.
- Brem, A., Radziwon, A., 2017. Efficient Triple Helix collaboration fostering local niche innovation projects-A case from Denmark. Technol. Forecast. Soc. Change 123, 130-141.
- Brown, R., Lee, N., 2019. Strapped for cash? Funding for UK high growth SMEs since the global financial crisis. J. Bus. Res. 99, 37-45.
- Bock, C., Huber, A., Jarchow, S., 2018. Growth factors of research-based spin-offs and the role of venture capital investing. J. Technol. Tran. 43 (5), 1375-1409.
- Bradley, S., Hayter, C.S., Link, A., 2013. Models and methods of university technology transfer. Foundations and trends in Entrepreneurship 9 (6).
- Burkholder, P., Hulsink, W., 2022. Academic intrapreneurship for health care innovation: the importance of influence, perception, and time management in

knowledge commercialization at a University's Medical Centre. J. Technol. Tran. 1-29.

- Cassiman, B., Veugelers, R., 2002. R&D cooperation and spillovers: some empirical evidence from Belgium. Am. Econ. Rev. 92 (4), 1169-1184.
- Chesbrough, H., Vanhaverbeke, W., West, J., 2014. New Frontiers in Open Innovation. Oup, Oxford.
- Chiesa, V., Piccaluga, A., 2000. Exploitation and diffusion of public research: the case of academic spin-off companies in Italy. R. Manag. 30 (4), 329-340.
- Churchill, N.C., Lewis, V.L., 1983. The five stages of small business growth. Harv. Bus. Rev. 61 (3), 30-50.
- Civera, A., Meoli, M., Vismara, S., 2020. Engagement of academics in university technology transfer: opportunity and necessity academic entrepreneurship. Eur. Econ. Rev. 123, 103376.
- Clarysse, B., Wright, M., Lockett, A., Mustar, P., Knockaert, M., 2007. Academic spin-offs, formal technology transfer and capital raising. Ind. Corp. Change 16 (4), 609-640. Clarysse, B., Wright, M., Lockett, A., Van de Velde, E., Vohora, A., 2005. Spinning out
- new ventures: a typology of incubation strategies from European research institutions. J. Bus. Ventur. 20 (2), 183-216.
- Clarysse, B., Wright, M., Van de Velde, E., 2011. Entrepreneurial origin, technological knowledge, and the growth of spin-off companies. J. Manag. Stud. 48 (6), 1420-1442
- Colombo, M., Mustar, P., Wright, M., 2010. Dynamics of science-based entrepreneurship. J. Technol. Tran. 35 (1), 1-15.
- Colombo, M.G., Meoli, M., Vismara, S., 2019. Signaling in science-based IPOs: the combined effect of affiliation with prestigious universities, underwriters, and venture capitalists. J. Bus. Ventur. 34 (1), 141-177.

Colombo, M.G., Montanaro, B., Vismara, S., 2022. What drives the valuation of entrepreneurial ventures? A map to navigate the literature and research directions. Small Bus. Econ. 1-26.

Covin, J.G., Slevin, D.P., 1991. A conceptual model of entrepreneurship as firm behaviour. Enterpren. Theor. Pract. 16 (1), 7-25.

- Connected Commerce Council, 2020. Digitally driven. U.S. Small businesses find a digital safety net during COVID-19 report. Available at: https://connectedcouncil.org/wp -content/uploads/2020/09/Digitally-Driven-Report.pdf. (Accessed 12 April 2021).
- Connected Commerce Council, 2021. Digitally Driven. European small business find a digital safety net during COVID-19. Available at: https://digitallydriven.connected council.org/europe/. (Accessed 13 April 2021).
- Cumming, D., Meoli, M., Vismara, S., 2021. Does equity crowdfunding democratize entrepreneurial finance? Small Bus. Econ. 56 (2), 533-552.
- Conti, A., Thursby, M., Rothaermel, F.T., 2013. Show me the right stuff: signals for hightech startups. J. Econ. Manag. Strat. 22 (2), 341-364.
- Cunningham, J.A., Dolan, B., Menter, M., O'Kane, C., O'Reilly, P., 2020. How principal investigators' commercial experience influences technology transfer and market impacts. Res. Technol. Manag. 63 (5), 49-58.
- Cunningham, J.A., Lehmann, E.E., Menter, M., 2022. The organizational architecture of entrepreneurial universities across the stages of entrepreneurship: a conceptual framework, Small Bus, Econ. 59 (1), 11-27.
- Di Gregorio, D., Shane, S., 2003. Why do some universities generate more start-ups than others? Res. Pol. 32 (2), 209-227.
- Druilhe, C., Garnsey, E., 2004. Do academic spin-outs differ and does it matter? J. Technol. Tran. 29 (3), 269-285.
- Farrukh, M., Chong, W.Y., Mansori, S., Ramzani, S.R., 2017. Intrapreneurial behaviour: the role of organizational commitment. World Journal of Entrepreneurship Management and Sustainable Development 13 (3), 243-256.
- Fukugawa, N., 2022. Effects of the quality of science and innovation on venture financing: evidence from university spinoffs in Japan. Appl. Econ. Lett. 1-7.
- Giones, F., Brem, A., 2017. Digital technology entrepreneurship: a definition and research agenda. Technology Innovation Management Review 7 (5).
- Gompers, P., 1995. Optimal investment, monitoring, and the staging of venture capital. J. Finance 50 (5), 1461-1489.
- Gonzalez, A., Lopez-Cordova, J.E., Valladares, E.E., 2012. The Incidence of Graft on Developing-Country Firms, vol. 4394. World Bank Publications.
- Gruber, M., MacMillan, I.C., Thompson, J.D., 2013. Escaping the prior knowledge corridor: what shapes the number and variety of market opportunities identified before market entry of technology start-ups? Organ. Sci. 24 (1), 280-300.
- Guerrero, M., Urbano, D., 2012. The development of an entrepreneurial university. J. Technol. Tran. 37 (1), 43-74.
- Guerrero, M., Peña-Legazkue, I., 2013. The effect of intrapreneurial experience on corporate venturing: evidence from developed economies. Int. Enterpren. Manag. J. 9 (3), 397-416.
- Guerrero, M., Urbano, D., 2014. Academics' start-up intentions and knowledge filters: an individual perspective of the knowledge spillover theory of entrepreneurship. Small Bus. Econ. 43 (1), 57-74.
- Guerrero, M., Cunningham, J.A., Urbano, D., 2015. Economic impact of entrepreneurial universities' activities: an exploratory study of the United Kingdom. Res. Pol. 44 (3), 748-764.
- Guerrero, M., Urbano, D., Fayolle, A., Klofsten, M., Mian, S., 2016. Entrepreneurial universities: emerging models in the new social and economic landscape. Small Bus. Econ. 47 (3), 551-563.
- Guerrero, M., Urbano, D., Gajón, E., 2020. Entrepreneurial university ecosystems and graduates' career patterns: do entrepreneurship education programmes and university business incubators matter? J. Manag. Dev. 39 (5), 753-775.
- Guerrero, M., Heaton, S., Urbano, D., 2021. Building universities' intrapreneurial capabilities in the digital era: the role and impacts of Massive Open Online Courses (MOOCs). Technovation 99, 102139.

Gustafsson, A., Tingvall, P.G., Halvarsson, D., 2020. Subsidy entrepreneurs: an inquiry into firms seeking public grants. J. Ind. Compet. Trade 20, 439–478.

Heblich, S., Slavtchev, V., 2014. Parent universities and the location of academic startups. Small Bus. Econ. 42, 1–15.

Heirman, A., Clarysse, B., 2004. How and why do research-based start-ups differ at founding? A resource-based configurational perspective. J. Technol. Tran. 29 (3), 247–268.

Jones-Evans, D., Klofsten, M., 1997. Universities and local economic development - the case of Linkoping, Sweden. Eur. Plann. Stud. 5 (1), 77–94.

Jovanovic, M., Sjödin, D., Parida, V., 2021. Co-evolution of Platform Architecture, Platform Services, and Platform Governance: Expanding the Platform Value of Industrial Digital Platforms. Technovation, 102218.

Klofsten, M., Jones-Evans, D., 2000. Comparing academic entrepreneurship in Europe-the case of Sweden and Ireland. Small Bus. Econ. 14 (4), 299–309. Klofsten, M., 2008. Supporting academic enterprise: a case study of an entrepreneurship

programme. N. Technol. Based Firms N. Millenium 6, 55–67. Klofsten, M., Fayolle, A., Guerrero, M., Mian, S., Urbano, D., Wright, M., 2019. The

entrepreneurial university as driver for economic growth and social change-Key strategic challenges. Technol. Forecast. Soc. Change 141, 149–158. Klofsten, M., Urbano, D., Heaton, S., 2021. Managing intrapreneurial capabilities: an

overview. Technovation 99, 102177.

Knockaert, M., Spithoven, A., Clarysse, B., 2010. The knowledge paradox explored: what is impeding the creation of ICT spin-offs? Technol. Anal. Strat. Manag. 22 (4), 479–493.

Knockaert, M., Ucbasaran, D., Wright, M., Clarysse, B., 2011. The relationship between knowledge transfer, top management team composition, and performance: the case of science-based entrepre-neurial firms. Entrep. Theory Pract. 35 (4), 777–803.

Lauto, G., Salvador, E., Visintin, F., 2022. For what they are, not for what they bring: the signaling value of gender for financial resource acquisition in academic spin-offs. Res. Pol. 51 (7), 104554.

Li, W., Liu, K., Belitski, M., Ghobadian, A., O'Regan, N., 2016. e-Leadership through strategic alignment: an empirical study of small-and medium-sized enterprises in the digital age. J. Inf. Technol. 31 (2), 185–206.

Lu, Q., Song, L., Yu, K., 2022. The effects of operational and digitalization capabilities on SMSs' supply chain financing under supply risk. Int. J. Logist. Res. Appl. 1–21.

Mariani, M.M., Belitski, M., 2022. The effect of coopetition intensity on first mover advantage and imitation in innovation related coopetition: empirical evidence from UK firms. Eur. Manag. J. https://doi.org/10.1016/j.emj.2022.05.001.

 Mathisen, M.T., Rasmussen, E., 2019. The development, growth, and performance of university spin-offs: a critical review. J. Technol. Tran. 44 (6), 1891–1938.
Meoli, M., Paleari, S., Vismara, S., 2019. The governance of universities and the

establishment of academic spin-offs. Small Bus. Econ. 52, 485–504.

Meoli, M., Vismara, S., 2016. University support and the creation of technology and nontechnology academic spin-offs. Small Bus. Econ. 47 (2), 345–362.

Moray, N., Clarysse, B., 2005. Institutional change and resource endowments to sciencebased entrepreneurial firms. Res. Pol. 34 (7), 1010–1027.

Munari, F., Toschi, L., 2011. Do venture capitalists have a bias against investment in academic spin-offs? Evidence from the micro-and nanotechnology sector in the UK. Ind. Corp. Change 20 (2), 397–432.

Mustar, P., Renault, M., Colombo, M.G., Piva, E., Fontes, M., Lockett, A., et al., 2006. Conceptualising the heterogeneity of research-based spin-offs: a multi-dimensional taxonomy. Res. Pol. 35 (2), 289–308.

Mustar, P., Wright, M., Clarysse, B., 2008. University spin-off firms: lessons from ten years of experience in Europe. Sci. Publ. Pol. 35 (2), 67–80.

Neessen, P.C., Caniëls, M.C., Vos, B., De Jong, J.P., 2019. The intrapreneurial employee: toward an integrated model of intrapreneurship and research agenda. Int. Enterpren. Manag. J. 15, 545–571.

OECD, 2015. Frascati Manual 2015: Guidelines for Collecting and Reporting Data on Research and Experimental Development, the Measurement of Scientific, Technological and Innovation Activities. OECD Publishing, Paris. https://doi.org/ 10.1787/978964239012-en Parker, S.C., 2011. Intrapreneurship or entrepreneurship? J. Bus. Ventur. 26 (1), 19–34. Perlines, F.H., Ariza-Montes, A., Blanco-González-Tejero, C., 2022. Intrapreneurship

research: a comprehensive literature review. J. Bus. Res. 153, 428–444. Pinchot III., G., 1985. Intrapreneuring: You Don't Have to Leave the Corporation to

- Become an Entrepreneur. Harper and Row, New York. Rabl, T., Petzsche, V., Baum, M., Franzke, S., 2022. Can support by digital technologies
- stimulate intrapreneurial behaviour? The moderating role of management support for innovation and intrapreneurial self-efficacy. Inf. Syst. J. 33 (3), 567–597.

Radko, N., Belitski, M., Kalyuzhnova, Y., 2022. Conceptualising the entrepreneurial university: the stakeholder approach. J. Technol. Tran. 1–90.

Rasmussen, E., Sørheim, R., 2012. How governments seek to bridge the financing gap for university spin-offs: proof-of-concept, pre-seed, and seed funding. Technol. Anal. Strat. Manag. 24 (7), 663–678.

Rippa, P., Secundo, G., 2019. Digital academic entrepreneurship: the potential of digital technologies on academic entrepreneurship. Technol. Forecast. Soc. Change 146, 900–911.

Rodríguez-Gulías, M.J., Rodeiro-Pazos, D., Fernández-López, S., Corsi, C., Prencipe, A., 2018. The role of venture capitalist to enhance the growth of Spanish and Italian university spin-offs. Int. Enterpren. Manag. J. 14, 1111–1130.

Secundo, G., Rippa, P., Cerchione, R., 2020. Digital Academic Entrepreneurship: a structured literature review and avenue for a research agenda. Technol. Forecast. Soc. Change 157, 120118.

Shane, S., 2004. Encouraging university entrepreneurship? The effect of the Bayh-Dole Act on university patenting in the United States. J. Bus. Ventur. 19 (1), 127–151.

Shane, S., Stuart, T., 2002. Organizational endowments and the performance of university start-ups. Manag. Sci. 48 (1), 154–170.

Soetanto, D., Van Geenhuizen, M., 2015. Getting the right balance: university networks' influence on spin-offs' attraction of funding for innovation. Technovation 36, 26–38.

Sørheim, R., Øystein Widding, L., Oust, M., Madsen, Ø., 2011. Funding of university spinoff companies: a conceptual approach to financing challenges. J. Small Bus. Enterprise Dev. 18 (1), 58–73.

Stuart, T.E., Ding, W.W., 2006. When do scientists become entrepreneurs? The social structural antecedents of commercial activity in the academic life sciences. Am. J. Sociol. 112, 97–144.

Teixeira, A.A., Ferreira, C., 2019. Intellectual property rights and the competitiveness of academic spin-offs. Journal of Innovation & Knowledge 4 (3), 154–161.

Unger, J.M., Rauch, A., Frese, M., Rosenbusch, N., 2011. Human capital and entrepreneurial success: a meta-analytical review. J. Bus. Ventur. 26 (3), 341–358. Urbano, D., Alvarez, C., Turró, A., 2013. Organizational resources and intrapreneurial

activities: an international study. Manag. Decis. 51 (4), 854–870. Vanhaverbeke, W., Peeters, N., 2005. Embracing innovation as strategy: corporate

vaniaverbeke, W., Peeters, N., 2005. Embracing innovation as strategy: corporate venturing, competence building and corporate strategy making. Creativ. Innovat. Manag. 14 (3), 246–257.

Valka, K., Roseira, C., Campos, P., 2020. Determinants of university employee intrapreneurial behavior: the case of Latvian universities. Ind. High. Educ. 34 (3), 190–202.

Van Burg, E., Romme, A.G.L., Gilsing, V.A., Reymen, I.M., 2008. Creating university spinoffs: a science-based design perspective. J. Prod. Innovat. Manag. 25 (2), 114–128.

Vincett, P.S., 2010. The economic impacts of academic spin-off companies, and their implications for public policy. Res. Pol. 39 (6), 736–747.

Visintin, F., Pittino, D., 2014. Founding team composition and early performance of university-based spinoff companies. Technovation 34 (1), 31–43.

Vohora, A., Wright, M., Lockett, A., 2004. Critical junctures in the development of university high-tech spinout companies. Res. Pol. 33 (1), 147–175.

Wooldridge, J.M., 2009. Introductory Economics: A Modern Approach. Mason, Cengage-Learning.

Wright, M., Lockett, A., Clarysse, B., Binks, M., 2006. University spin-out companies and venture capital. Res. Pol. 35 (4), 481–501.

Zhang, T., Gerlowski, D., Acs, Z., 2021. Working from home: small business performance and the COVID-19 pandemic. Small Bus. Econ. 1–26.